Author's response to reviews

Title: Matrix metalloproteinases and tissue inhibitors of metalloproteinases in the aqueous humour of patients with primary angle closure glaucoma - a quantitative study

Authors:

    Angeline DC Nga (angelinenga@yahoo.com)
    Yap Soon-Lek (soonlek@hotmail.com)
    Amir Samsudin (amir@ummc.edu.my)
    Puteri S Abdul-Rahman (terisar@um.edu.my)
    Onn H Hashim (onnhashim@um.edu.my)
    Zahari Mimiwati (mimiwati@ummc.edu.my)

Version: 4 Date: 7 March 2014

Author's response to reviews: see over
Dear Editor,

Thank you for pointing out the typographical error in our article. The definition mentioned by the Associate Editor was indeed what we used in our study. The text has now been corrected. Thank you again.

Yours sincerely,

Amir Samsudin

(on behalf of all the authors)